Author's response to reviews

Title: Shoshin Beriberi in Critically-Ill Patients: Case Series

Authors:

George R Dabar (georgedabar@yahoo.com)
Carine R Harmouche (kareenhr@hotmail.com)
Bassem M Habr (bassem.habr@gmail.com)
Moussa A Riachi (riachy@dm.net.lb)
Bertrand L Jaber (bertrand.jaber@steward.org)

Version: 3 Date: 22 April 2015

Author's response to reviews: see over
Dear Editor,

We are pleased to resubmit our manuscript entitled “Shoshin Beriberi in Critically-Ill Patients: A Case Series”, for consideration in your honorable journal, Case Reports section.

The comments forwarded by your reviewers were taken into consideration and the point-by-point response to the concerns is addressed.

Ref#2: Correct the page numbers: Done

Ref#4, 5, 12,14,16,18,20: Provide the full-page numbers, if it is the format followed throughout: Done

Ref#13,15,19: Are these single page articles? Yes, those are single page articles.

Figures 1A to 1D: Label X – and Y- axis. Labeled

Figure 2: Provide a very short description of this figure as a figure legend. The legend is expanded to provide a short description of figure2.

We look forward to see our publication in your journal.

Sincerely,

George R. Dabar MD
Assistant Professor of Medicine
Hotel Dieu de France Hospital
Saint Joseph University
PO Box 16-6830, Achrafieh
Lebanon
e-mail: georges.dabar@usj.edu.lb
georgedabar@yahoo.com